\documentclass{article}
%To use pdflatex, uncomment these lines, as well as the \href lines
%in each entry
%\usepackage[pdftex,
%       colorlinks=true,
%       urlcolor=blue,       % \href{...}{...} external (URL)
%       filecolor=green,     % \href{...} local file
%       linkcolor=red,       % \ref{...} and \pageref{...}
%       pdftitle={Papers by AUTHOR},
%       pdfauthor={Your Name},
%       pdfsubject={Just a test},
%       pdfkeywords={test testing testable},
%       pagebackref,
%       pdfpagemode=None,
%        bookmarksopen=true]{hyperref}
%usepackage{arial}
%\renewcommand{\familydefault}{\sfdefault} % San serif

\pagestyle{empty} \oddsidemargin 0.0in \textwidth 6.5in \topmargin
-0.75in \textheight 9.5in

\begin{document}
\title{Papers by Andrew C. Haas}
\author{}
\date{}
\maketitle
\begin{enumerate}


%%   LIST OF PAPERS
%%   Please comment out anything between here and the
%%   first \item
%%   Please send any updates or corrections to the list to
%%   library@slac.stanford.edu

%\cite{Abazov:2006xq}
\item%{Abazov:2006xq}
{\bf ``Limits on anomalous trilinear gauge couplings from $W W \to
e^+ e^-$, $W W \to e^\pm \mu^\mp$, and $W W \to \mu^+ \mu^-$ events
from $p \bar{p}$ collisions at $\sqrt{s}$ = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ D {\bf 74}, 057101 (2006)
  [arXiv:hep-ex/0608011]
%\\{}FERMILAB-PUB-06-268-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0608011}{SPIRES entry}

%\cite{Abazov:2006wp}
\item%{Abazov:2006wp}
{\bf ``Search for scalar leptoquarks in the acoplanar jet topology
in p anti-p collisions at s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 640}, 230 (2006)
  [arXiv:hep-ex/0607009]
%\\{}FERMILAB-PUB-06-233-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0607009}{SPIRES entry}

%\cite{Abazov:2006nr}
\item%{Abazov:2006nr}
{\bf ``Search for a heavy resonance decaying into a Z + jet final
state in p anti-p collisions at s**(1/2) = 1.96-TeV using the D0
detector''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ D {\bf 74}, 011104 (2006)
  [arXiv:hep-ex/0606018]
%\\{}FERMILAB-PUB-06-167-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0606018}{SPIRES entry}

%\cite{Abazov:2006ii}
\item%{Abazov:2006ii}
{\bf ``Search for resonant second generation slepton production at
the Tevatron''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 97}, 111801 (2006)
  [arXiv:hep-ex/0605010]
%\\{}FERMILAB-PUB-06-094-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0605010}{SPIRES entry}

%\cite{Abazov:2006ih}
\item%{Abazov:2006ih}
{\bf ``Search for neutral Higgs bosons decaying to tau pairs in p
anti-p collisions at s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 97}, 121802 (2006)
  [arXiv:hep-ex/0605009]
%\\{}FERMILAB-PUB-06-092-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0605009}{SPIRES entry}

%\cite{Abazov:2006nw}
\item%{Abazov:2006nw}
{\bf ``Search for R-parity violating supersymmetry via the LLE
couplings lambda(121), lambda(122) or lambda(133) in p anti-p
collisions at s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 638}, 441 (2006)
  [arXiv:hep-ex/0605005]
%\\{}FERMILAB-PUB-06-089-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0605005}{SPIRES entry}

%\cite{Abazov:2006cb}
\item%{Abazov:2006cb}
{\bf ``A precise measurement of the B/s0 lifetime''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}arXiv:hep-ex/0604046
%\\{}FERMILAB-PUB-06-085-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0604046}{SPIRES entry}
(Submitted to Phys.Rev.Lett.)

%\cite{Abazov:2006wq}
\item%{Abazov:2006wq}
{\bf ``Search for excited muons in p anti-p collisions at s**(1/2) =
1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ D {\bf 73}, 111102 (2006)
  [arXiv:hep-ex/0604040]
%\\{}FERMILAB-PUB-06-081-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0604040}{SPIRES entry}

%\cite{Abazov:2006bj}
\item%{Abazov:2006bj}
{\bf ``Search for squarks and gluinos in events with jets and
missing transverse energy in p anti-p collisions at s**(1/2) =
1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 638}, 119 (2006)
  [arXiv:hep-ex/0604029]
%\\{}FERMILAB-PUB-06-077-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0604029}{SPIRES entry}

%\cite{Abazov:2006uq}
\item%{Abazov:2006uq}
{\bf ``Multivariate searches for single top quark production with
the D0 detector''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}arXiv:hep-ex/0604020
%\\{}FERMILAB-PUB-06-069-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0604020}{SPIRES entry}
(Submitted to Phys.Rev.D)

%\cite{Abazov:2006qm}
\item%{Abazov:2006qm}
{\bf ``Search for the rare decay B0/s --> Phi mu+ mu- with the D0
detector''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ D {\bf 74}, 031107 (2006)
  [arXiv:hep-ex/0604015]
%\\{}FERMILAB-PUB-06-073-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0604015}{SPIRES entry}

%\cite{Abazov:2006dm}
\item%{Abazov:2006dm}
{\bf ``First direct two-sided bound on the B/s0 oscillation
frequency''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 97}, 021802 (2006)
  [arXiv:hep-ex/0603029]
%\\{}FERMILAB-PUB-06-055-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0603029}{SPIRES entry}

%\cite{Abazov:2006bh}
\item%{Abazov:2006bh}
{\bf ``Measurement of B(t --> W b)/B(t --> W q) at s**(1/2) =
1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 639}, 616 (2006)
  [arXiv:hep-ex/0603002]
%\\{}FERMILAB-PUB-06-037-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0603002}{SPIRES entry}

%\cite{Abazov:2006vc}
\item%{Abazov:2006vc}
{\bf ``Search for pair production of second generation scalar
leptoquarks in p  anti-p collisions at s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 636}, 183 (2006)
  [arXiv:hep-ex/0601047]
%\\{}FERMILAB-PUB-06-017-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0601047}{SPIRES entry}

%\cite{Abazov:2005wc}
\item%{Abazov:2005wc}
{\bf ``Measurement of the isolated photon cross section in p anti-p
collisions  at s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 639}, 151 (2006)
  [arXiv:hep-ex/0511054]
%\\{}FERMILAB-PUB-05-523-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0511054}{SPIRES entry}

%\cite{Abazov:2005un}
\item%{Abazov:2005un}
{\bf ``Search for the Higgs boson in H --> W W(*) decays in p anti-p
collisions at s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 96}, 011801 (2006)
  [arXiv:hep-ex/0508054]
%\\{}FERMILAB-PUB-05-377-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0508054}{SPIRES entry}

%\cite{Abazov:2005pn}
\item%{Abazov:2005pn}
{\bf ``The upgraded D0 detector''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Nucl.\ Instrum.\ Meth.\ A {\bf 565}, 463 (2006)
  [arXiv:physics/0507191]
%\\{}FERMILAB-PUB-05-341-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=physics\%2F0507191}{SPIRES entry}

%\cite{Abazov:2005sa}
\item%{Abazov:2005sa}
{\bf ``Measurement of the lifetime difference in the B/s0 system''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 95}, 171801 (2005)
  [arXiv:hep-ex/0507084]
%\\{}FERMILAB-PUB-05-324-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0507084}{SPIRES entry}

%\cite{Abazov:2005ga}
\item%{Abazov:2005ga}
{\bf ``Measurement of semileptonic branching fractions of B mesons
to narrow  D** states''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 95}, 171803 (2005)
  [arXiv:hep-ex/0507046]
%\\{}FERMILAB-PUB-05-313-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0507046}{SPIRES entry}

%\cite{Abazov:2005tk}
\item%{Abazov:2005tk}
{\bf ``Search for large extra spatial dimensions in dimuon
production at D0''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 95}, 161602 (2005)
  [arXiv:hep-ex/0506063]
%\\{}FERMILAB-PUB-05-250-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0506063}{SPIRES entry}

%\cite{Abazov:2005yt}
\item%{Abazov:2005yt}
{\bf ``Measurement of the t anti-t production cross section in p
anti-p  collisions at s**(1/2) = 1.96-TeV in dilepton final
states''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 626}, 55 (2005)
  [arXiv:hep-ex/0505082]
%\\{}FERMILAB-PUB-05-217-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0505082}{SPIRES entry}

%\cite{Abazov:2005zz}
\item%{Abazov:2005zz}
{\bf ``Search for single top quark production in p anti-p collisions
at  s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 622}, 265 (2005)
  [arXiv:hep-ex/0505063]
%\\{}FERMILAB-PUB-05-207-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0505063}{SPIRES entry}

%\cite{Abazov:2005fk}
\item%{Abazov:2005fk}
{\bf ``Measurement of the W boson helicity in top quark decays''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ D {\bf 72}, 011104 (2005)
  [arXiv:hep-ex/0505031]
%\\{}FERMILAB-PUB-05-187-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0505031}{SPIRES entry}

%\cite{Abazov:2005pi}
\item%{Abazov:2005pi}
{\bf ``Search for Randall-Sundrum gravitons in dilepton and diphoton
final  states''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 95}, 091801 (2005)
  [arXiv:hep-ex/0505018]
%\\{}FERMILAB-PUB-05-126-E-T
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0505018}{SPIRES entry}

%\cite{Abazov:2005ey}
\item%{Abazov:2005ey}
{\bf ``Measurement of the t anti-t production cross section in p
anti-p  collisions at s**(1/2) = 1.96-TeV using lepton + jets events
with lifetime b-tagging''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 626}, 35 (2005)
  [arXiv:hep-ex/0504058]
%\\{}FERMILAB-PUB-05-087-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0504058}{SPIRES entry}

%\cite{Abazov:2005ex}
\item%{Abazov:2005ex}
{\bf ``Measurement of the t anti-t production cross section in p
anti-p  collisions at s**(1/2) = 1.96-TeV using kinematic
characteristics of lepton + jets events''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Lett.\ B {\bf 626}, 45 (2005)
  [arXiv:hep-ex/0504043]
%\\{}FERMILAB-PUB-05-079-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0504043}{SPIRES entry}

%\cite{Abazov:2005ku}
\item%{Abazov:2005ku}
{\bf ``Search for supersymmetry via associated production of
charginos and  neutralinos in final states with three leptons''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 95}, 151805 (2005)
  [arXiv:hep-ex/0504032]
%\\{}FERMILAB-PUB-05-075-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0504032}{SPIRES entry}

%\cite{Abazov:2005ys}
\item%{Abazov:2005ys}
{\bf ``Production of W Z events in p anti-p collisions at s**(1/2) =
1.96-TeV  and limits on anomalous W W Z couplings''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 95}, 141802 (2005)
  [arXiv:hep-ex/0504019]
%\\{}FERMILAB-PUB-05-061-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0504019}{SPIRES entry}

%\cite{Abazov:2005yr}
\item%{Abazov:2005yr}
{\bf ``Search for neutral supersymmetric Higgs bosons in multijet
events at  s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 95}, 151801 (2005)
  [arXiv:hep-ex/0504018]
%\\{}FERMILAB-PUB-05-058-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0504018}{SPIRES entry}

%\cite{Abazov:2005ni}
\item%{Abazov:2005ni}
{\bf ``Measurement of the p anti-p --> W gamma + X cross section at
s**(1/2) =  1.96-TeV and W W gamma anomalous coupling limits''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ D {\bf 71}, 091108 (2005)
  [arXiv:hep-ex/0503048]
%\\{}FERMILAB-PUB-05-046-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0503048}{SPIRES entry}

%\cite{Abazov:2005ea}
\item%{Abazov:2005ea}
{\bf ``Study of Z gamma events and limits on anomalous Z Z gamma and
Z gamma  gamma couplings in p anti-p collisions at s**(1/2) =
1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 95}, 051802 (2005)
  [arXiv:hep-ex/0502036]
%\\{}FERMILAB-PUB-05-023-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0502036}{SPIRES entry}

%\cite{Abazov:2005yc}
\item%{Abazov:2005yc}
{\bf ``Measurement of inclusive differential cross sections for
Upsilon(1S)  production in p anti-p collisions at s**(1/2) =
1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 232001 (2005)
  [arXiv:hep-ex/0502030]
%\\{}FERMILAB-PUB-05-020-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0502030}{SPIRES entry}

%\cite{Angstadt:2004bi}
\item%{Angstadt:2004bi}
{\bf ``The DZERO level 3 data acquisition system''}
  \\{}R.~D.~Angstadt {\it et al.}
  \\{}IEEE Trans.\ Nucl.\ Sci.\  {\bf 51}, 445 (2004)
%\href{http://www.slac.stanford.edu/spires/find/hep/www?j=ietna\%2c51\%2c445}{SPIRES entry}

%\cite{Abazov:2004mk}
\item%{Abazov:2004mk}
{\bf ``Search for first-generation scalar leptoquarks in p anti-p
collisions  at s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ D {\bf 71}, 071104 (2005)
  [arXiv:hep-ex/0412029]
%\\{}FERMILAB-PUB-04-389-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0412029}{SPIRES entry}

%\cite{Abazov:2004vd}
\item%{Abazov:2004vd}
{\bf ``First measurement of sigma(p anti-p --> Z) x Br(Z --> tau
tau) at  s**(1/2) = 1.96-TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ D {\bf 71}, 072004 (2005)
  [arXiv:hep-ex/0412020]
%\\{}FERMILAB-PUB-04-381-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0412020}{SPIRES entry}

%\cite{Abazov:2004bv}
\item%{Abazov:2004bv}
{\bf ``A search for anomalous heavy-flavor quark production in
association  with W bosons''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 152002 (2005)
  [arXiv:hep-ex/0411084]
%\\{}FERMILAB-PUB-04-359-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0411084}{SPIRES entry}

%\cite{Abazov:2004zd}
\item%{Abazov:2004zd}
{\bf ``A measurement of the ratio of inclusive cross sections
$\sigma(p p \to Z + b-{\rm jet})/\sigma(p p \to Z + jet)$ at
$\sqrt{s} = 1.96$ TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 161801 (2005)
  [arXiv:hep-ex/0410078]
%\\{}FERMILAB-PUB-04-297-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0410078}{SPIRES entry}

%\cite{Abazov:2004kc}
\item%{Abazov:2004kc}
{\bf ``Measurement of the $W W$ production cross section in
$p\bar{p}$ collisions at $\sqrt{s} = 1.96$ TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 151801 (2005)
  [arXiv:hep-ex/0410066]
%\\{}FERMILAB-PUB-04-293-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0410066}{SPIRES entry}

%\cite{Abazov:2004jy}
\item%{Abazov:2004jy}
{\bf ``A search for $W b\bar{b}$ and $W H$ production in $p\bar{p}$
collisions at $\sqrt{s} = 1.96$ TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 091802 (2005)
  [arXiv:hep-ex/0410062]
%\\{}FERMILAB-PUB-04-288-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0410062}{SPIRES entry}

%\cite{Abazov:2004bn}
\item%{Abazov:2004bn}
{\bf ``Measurement of the $\Lambda_b^0$ lifetime in the decay
$\Lambda_b^0 \to J/\psi \Lambda^0$ with the D\O\ detector''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 102001 (2005)
  [arXiv:hep-ex/0410054]
%\\{}FERMILAB-PUB-04-286-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0410054}{SPIRES entry}

%\cite{Abazov:2004sa}
\item%{Abazov:2004sa}
{\bf ``Measurement of the ratio of $B^+$ and $B^0$ meson
lifetimes''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 182001 (2005)
  [arXiv:hep-ex/0410052]
%\\{}FERMILAB-PUB-04-284-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0410052}{SPIRES entry}

%\cite{Abazov:2004dj}
\item%{Abazov:2004dj}
{\bf ``A search for the flavor-changing neutral current decay $B_s^0
\to \mu^+ \mu^-$ in $p\bar{p}$ collisions at $\sqrt{s} = 1.96$ TeV
with the D\O\ detector''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 071802 (2005)
  [arXiv:hep-ex/0410039]
%\\{}FERMILAB-PUB-04-215-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0410039}{SPIRES entry}

%\cite{Abazov:2004ce}
\item%{Abazov:2004ce}
{\bf ``Measurement of the $B_s^0$ lifetime in the exclusive decay
channel $B_s^0 \to J/\psi \phi$''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 042001 (2005)
  [arXiv:hep-ex/0409043]
%\\{}FERMILAB-PUB-04-225-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0409043}{SPIRES entry}

%\cite{Abazov:2004hm}
\item%{Abazov:2004hm}
{\bf ``Measurement of dijet azimuthal decorrelations at central
rapidities in $p\bar{p}$ collisions at $\sqrt{s} = 1.96$ TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 221801 (2005)
  [arXiv:hep-ex/0409040]
%\\{}FERMILAB-PUB-04-217-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0409040}{SPIRES entry}

%\cite{Abazov:2004jx}
\item%{Abazov:2004jx}
{\bf ``Search for supersymmetry with gauge-mediated breaking in
diphoton events at D\O''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 94}, 041801 (2005)
  [arXiv:hep-ex/0408146]
%\\{}FERMILAB-PUB-04-198-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0408146}{SPIRES entry}

%\cite{Abazov:2004kp}
\item%{Abazov:2004kp}
{\bf ``Observation and properties of the $X(3872)$ decaying to
$J/\psi \pi^+ \pi^-$ in $p\bar{p}$ collisions at $\sqrt{s} = 1.96$
TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 93}, 162002 (2004)
  [arXiv:hep-ex/0405004]
%\\{}FERMILAB-PUB-04-061-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0405004}{SPIRES entry}

%\cite{Abazov:2004au}
\item%{Abazov:2004au}
{\bf ``Search for doubly-charged Higgs boson pair production in the
decay to $\mu^+ \mu^+ \mu^- \mu^-$ in $p\bar{p}$ collisions at
$\sqrt{s} = 1.96$ TeV''}
  \\{}V.~M.~Abazov {\it et al.}  [D0 Collaboration]
  \\{}Phys.\ Rev.\ Lett.\  {\bf 93}, 141801 (2004)
  [arXiv:hep-ex/0404015]
%\\{}FERMILAB-PUB-04-045-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=hep-ex\%2F0404015}{SPIRES entry}

%\cite{Haas:2004rx}
\item%{Haas:2004rx}
{\bf ``A search for neutral Higgs bosons at high tan beta in
multi-jet events  from p anti-p collisions at s**(1/2) = 1960-GeV''}
  \\{}A.~C.~Haas
\\{}FERMILAB-THESIS-2004-26
%\href{http://www.slac.stanford.edu/spires/find/hep/www?r=fermilab-thesis-2004-26}{SPIRES entry}

%\cite{Chapin:2003xc}
\item%{Chapin:2003xc}
{\bf ``The DZERO level 3 data acquisition system''}
  \\{}D.~Chapin {\it et al.}
  \\{}eConf {\bf C0303241}, TUGP010 (2003)
  [eConf {\bf C0303241}, MOGT002 (2003)]
  [arXiv:physics/0307070]
%\\{}FERMILAB-CONF-03-462-E
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=physics\%2F0307070}{SPIRES entry}

%\cite{Haas:2003dy}
\item%{Haas:2003dy}
{\bf ``D0 online monitoring and automatic DAQ recovery''}
  \\{}A.~Haas {\it et al.}
  \\{}eConf {\bf C0303241}, TUGP011 (2003)
  [eConf {\bf C0303241}, THGT004 (2003)]
  [arXiv:physics/0306195]
%\\{}FERMILAB-CONF-03-467
%\href{http://www.slac.stanford.edu/spires/find/hep/www?eprint=physics\%2F0306195}{SPIRES entry}

%\cite{Haas:2000xj}
\item%{Haas:2000xj}
{\bf ``Simultaneous tracking and vertexing with elastic templates''}
  \\{}A.~Haas
%\href{http://www.slac.stanford.edu/spires/find/hep/www?irn=4889622}{SPIRES entry}
\\{}{\it Prepared for 7th International Workshop on Advanced Computing and Analysis Techniques in Physics Research (ACAT 2000), Batavia,
Illinois, 16-20 Oct 2000}


\end{enumerate}
\end{document}
